Summary of Installation Activities: 3/23/2006
L1 Calorimeter Trigger Upgrade

1. Some more infrastructural work asked by Mike and Dan are done. Some of the blanks are in place.
2. The cables on the third panel of M110 were pulled. All is well.

3. I finished cabling M110. Dan and I did the pulser test. The results are on the attached excel file (also on web). Four cables were shorted. Dan will fix them tonight. Others have BLS cable/hybrid problems, according to Dan. We will discuss this with Norm in the morning. Dean may have comments also.

4. On the attached pulser data excel file the results for M112 are still preliminary. I will redo the test with signal values as data. Dean commented on the results re-iterating that the results are hard to interpret on this rack (eta-phi) region.
Please see the web for details. URL is http://d0server1.fnal.gov/users/selcuk/www/default.htm. This information is on the trigger online logbook also.
Selcuk Cihangir
